Author's response to reviews

Title: Frequency Format Diagram and Probability Chart for Breast Cancer Risk Communication: a Prospective, Randomized Trial

Authors:

Karthik Ghosh (ghosh.karthik@mayo.edu)
Brianna J Crawford (crawford.brianna@mayo.edu)
Sandhya Pruthi (pruthi.sandhya@mayo.edu)
Constance I Williams (williams.constance@mayo.edu)
Lonzetta Neal (neal.lonzetta@mayo.edu)
Nicole P Sandhu (sandhu.nicole@mayo.edu)
Ruth E Johnson (johnsonmd.ruth@mayo.edu)
Dietlind Wahner-Roedler (wahnerroedler.dietlind@mayo.edu)
Marcia K Britain (britain.marcia@mayo.edu)
Stephen S Cha (cha@mayo.edu)
Amit K Ghosh (ghosh.amit@mayo.edu)

Version: 6 Date: 14 October 2008

Author's response to reviews: see over
October 14, 2008

The Biomed Central Editorial Production Team
34-42 Cleveland Street
London W1T 4LB
United Kingdom

Subject: Formatting Changes for MS: 1656132309189872

Thank you for your suggestions. We have made the necessary formatting changes as noted below:

1) Author list - The list of authors in the manuscript should be written exactly as they are in the submission system.

Completed as suggested.
Does each author have to be listed on separate lines as we had done previously?
Please let me know if further changes are needed - thanks.

2) Email address - please include email addresses for all the authors on the title page.

Email addresses were included on the title page.

3) Please remove the Table and Figure counts.

Table and figure counts deleted from the title page.

4) Previous exhibition - If you have previously exhibited your work at a conference (or similar), details of this should be provided in the Acknowledgements only.

Please remove this statement from the Abstract.

This section was moved to the Acknowledgments section.

5) Keywords - Please remove these.

Keywords were deleted.

Sincerely,

Karthik

Karthik Ghosh MD, MS
Assistant Professor
Breast Diagnostic Clinic
Division of General Internal Medicine
Mayo Clinic
Rochester, MN 55905
Phone: 507-284-9039
Fax: 507-284-4959